Author's response to reviews

Title: The use of annual physical examination by rural older adults in China: a cross-sectional study

Authors:

Xi SUN (sunxi88543470@gmai.com)
Xuetao TONG (1115393686@qq.com)
Zhanchun FENG (zcfeng@hust.edu.cn)
HoTing WONG (frankwong@connect.hku.hk)
li WEI (65943390@qq.com)
Donghua ZHOU (357597090@qq.com)
Miaomiao TIAN (283025868@qq.com)
Benyan LV (10455021@qq.com)
Da FENG (857801173@qq.com)

Version: 2 Date: 13 March 2013

Author's response to reviews:

Dear Editor,
Thank you for your suggestion.
The abstract has been revised.
With best wishes,
Zhanchun Feng
email:zcfeng@hust.edu.cn